Author's response to reviews

Title: Evaluation of electronic patient-reported outcome assessment with cancer patients in the hospital and at home

Authors:

Lisa M Wintner (lisa.wintner@i-med.ac.at)
Johannes M Giesinger (johannes.giesinger@i-med.ac.at)
August Zabernigg (august.zabernigg@bkh-kufstein.at)
Gerhard Rumpold (gerhard.rumpold@tirol-kliniken.at)
Monika Sztankay (monika.sztankay@tirol-kliniken.at)
Anne S Oberguggenberger (anne.oberguggenberger@tirol-kliniken.at)
Eva-Maria Gamper (eva.gamper@tirol-kliniken.at)
Bernhard Holzner (bernhard.holzner@tirol-kliniken.at)

Version: 6 Date: 25 November 2015

Author's response to reviews: see over
Dear editor,

please find enclosed our manuscript entitled “Evaluation of electronic patient-reported outcome assessment with cancer patients in the hospital and at home”, which we would like to resubmit after revision as an original article for publication in BMC Medical Informatics & Decision Making.

We thank you for your comments and would like to give a reply to each of them:

1. Authorship: We note that since the original submission, Gerhard Rumpold been added to the list of authors. We must therefore require that all authors, Gerhard Rumpold included, individually send us an email expressing that they approve the current list of authors.

All authors have already be informed about the current list of authors and will, within the next few days, provide the requested email expressing their approval of it.

2. Consent: Please specify if the consent the participants provided was written or verbal.

We included the information that patients gave written informed consent in the methods section (sample and procedure, p. 5-6, lines 101-111).

We hope that the response meets the editor’s expectations, with the result that our paper now complies with the requirements for publication.

Again, we confirm that neither the submitted material nor portions thereof have been published previously or are under consideration for publication elsewhere. The revised version of the article has been approved by all authors.

We are looking forward to hearing from you.

Yours truly,

Lisa M. Wintner, Msc

Bernhard Holzner, PhD.Prof.